Hall D Controls Meeting 01-Nov-2018

Date: 11/01/18 Time: 10:00am

<u>Attendees</u>: Peter Bonneau, Pablo Campero, Hovanes Egiyan, Brian Eng, Amanda Hoebel, Tyler Lemon, Nick Sandoval, Tim Whitlatch

1. General:

1.1. Beam will run until Nov 26.

2. ComCal:

- 2.1. There will be 6 days of downtime starting Nov 26 for installation.
 - 2.1.1. ComCal will be moved to stage, cabled, and tested.
- 2.2. GUI being constructed.
 - 2.2.1. Has darkbox temperatures, light sensor values, crystal temperatures, and HV interlock.
- 2.3. Interlocks tested and works.
- 2.4. LED pulser was not working.
 - 2.4.1. Problem found to be incorrect board address.

3. DIRC controls:

- 3.1. DIRC to be installed in January.
- 3.2. IOCs and GUIs completed.
 - 3.2.1. HV cannot be turned on or off.
- 3.3. Do not have control board for LED pulser.